Author’s response to reviews

Title: The impact of cancer on diabetes outcomes.

Authors:
Anne Arreskov (anne.arreskov@sund.ku.dk)
Maria Olsen (maria@aahlander.dk)
Sandra Pouplier (sandra.pouplier@gmail.com)
Volkert Siersma (siersma@sund.ku.dk)
Christen Andersen (christenla@gmail.com)
Søren Friis (friis@CANCER.DK)
Niels Olivarius (olivarius@sund.ku.dk)

Version: 2 Date: 17 Apr 2019

Author’s response to reviews:

Dear Editor and reviewers

Thank you very much for having read our article again. We have now worked thoroughly through the manuscript and revised it according to the comments.

We have attached a “clean version” of the manuscript: “Main Document_R2_clean”.

We look forward to hearing from you again.

On behalf of all authors.

Yours sincerely

Anne Arreskov

M.D. Ph.D student
Research Unit for General Practice in Copenhagen
University of Copenhagen